

Predicted chance that global warming will temporarily exceed 1.5 °C

Article

Published Version

Creative Commons: Attribution-Noncommercial-No Derivative Works 4.0

Open access

Smith, D. M., Scaife, A. A., Hawkins, E. ORCID: https://orcid.org/0000-0001-9477-3677, Bilbao, R., Boer, G. J., Caian, M., Caron, L.-P., Danabasoglu, G., Delworth, T., Doblas-Reyes, F. J., Doescher, R., Dunstone, N. J., Eade, R., Hermanson, L., Ishii, M., Kharin, V., Kimoto, M., Koenigk, T., Kushnir, Y., Matei, D., Meehl, G. A., Menegoz, M., Merryfield, W. J., Mochizuki, T., Müller, W. A., Pohlmann, H., Power, S., Rixen, M., Sospedra-Alfonso, R., Tuma, M., Wyser, K., Yang, X. and Yeager, S. (2018) Predicted chance that global warming will temporarily exceed 1.5 °C. Geophysical Research Letters, 45 (21). pp. 11895-11903. ISSN 0094-8276 doi: https://doi.org/10.1029/2018GL079362 Available at https://centaur.reading.ac.uk/80380/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1029/2018GL079362

Publisher: American Geophysical Union

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other



copyright holders. Terms and conditions for use of this material are defined in the End User Agreement.

www.reading.ac.uk/centaur

CentAUR

Central Archive at the University of Reading

Reading's research outputs online



Geophysical Research Letters

RESEARCH LETTER

10.1029/2018GL079362

Key Points:

- Early temporary excursions above 1.5 °C would provide a warning that one of the Paris Agreement thresholds is being approached
- Initialized climate predictions indicate a 38% (10%) chance of at least 1 month (year) exceeding 1.5 °C in the 5 year period 2017-2021
- Five-year mean temperatures above 1.5 °C are extremely unlikely in this period

Correspondence to:

D. M. Smith, doug.smith@metoffice.gov.uk

Citation:

Smith, D. M., Scaife, A. A., Hawkins, E., Bilbao, R., Boer, G. J., Caian, M., et al. (2018). Predicted chance that global warming will temporarily exceed 1.5 °C. *Geophysical Research Letters*, 45. https:// doi.org/10.1029/2018GL079362

Received 26 JUN 2018 Accepted 8 OCT 2018 Accepted article online 12 OCT 2018

©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Predicted Chance That Global Warming Will Temporarily Exceed 1.5 °C

D. M. Smith¹ (D, A. A. Scaife^{1,2} (D, E. Hawkins³ (D, R. Bilbao⁴ (D, G. J. Boer⁵, M. Caian⁶, L.-P. Caron⁴ (D, G. Danabasoglu⁷ (D, T. Delworth⁸ (D, F. J. Doblas-Reyes^{4,9}, R. Doescher⁶, N. J. Dunstone¹ (D, R. Eade¹ (D, L. Hermanson¹ (D, M. Ishii¹⁰ (D, V. Kharin⁵ (D, M. Kimoto¹¹ (D, T. Koenigk⁶ (D, Y. Kushnir¹² (D, D. Matei¹³, G. A. Meehl⁷ (D, M. Menegoz⁴ (D, W. J. Merryfield⁵ (D, T. Mochizuki¹⁴ (D, W. A. Müller^{13,15} (D, H. Pohlmann¹³ (D, S. Power¹⁶ (D, M. Rixen¹⁷ (D, R. Sospedra-Alfonso⁵, M. Tuma¹⁷ (D, K. Wyser⁶ (D, X. Yang⁸ (D, and S. Yeager⁷ (D))

¹Met Office Hadley Centre, FitzRoy Road, Exeter, UK, ²College of Engineering, Mathematics and Physical Sciences, University of Exeter, Exeter, UK, ³NCAS-Climate, Department of Meteorology, University of Reading, Reading, UK, ⁴Barcelona Supercomputing Center, Barcelona, Spain, ⁵Canadian Centre for Climate Modelling and Analysis, Environment and Climate Change Canada, Victoria, British Columbia, Canada, ⁶Rossby Centre, Swedish Meteorological and Hydrological Institute, Norrköping, Sweden, ⁷National Center for Atmospheric Research, Boulder, CO, USA, ⁸Geophysical Fluid Dynamics Laboratory, Princeton University, Princeton, NJ, USA, ⁹ICREA, Barcelona, Spain, ¹⁰Meteorological Research Institute, Japan Meteorological Agency, Tsukuba, Japan, ¹¹Atmosphere and Ocean Research Institute, University of Tokyo, Kashiwa, Japan, ¹²Lamont-Doherty Earth Observatory, Earth Institute, Columbia University, New York, NY, USA, ¹³Max-Planck-Institut für Meteorologie, Hamburg, Germany, ¹⁴Japan Agency for Marine-Earth Science and Technology, Yokohama, Japan, ¹⁵Deutscher Wetterdienst, Hamburg, Germany, ¹⁶Bureau of Meteorology, Melbourne, Australia, ¹⁷WCRP/WMO, Geneva, Switzerland

Abstract The Paris Agreement calls for efforts to limit anthropogenic global warming to less than 1.5 °C above preindustrial levels. However, natural internal variability may exacerbate anthropogenic warming to produce temporary excursions above 1.5 °C. Such excursions would not necessarily exceed the Paris Agreement, but would provide a warning that the threshold is being approached. Here we develop a new capability to predict the probability that global temperature will exceed 1.5 °C above preindustrial levels in the coming 5 years. For the period 2017 to 2021 we predict a 38% and 10% chance, respectively, of monthly or yearly temperatures exceeding 1.5 °C, with virtually no chance of the 5-year mean being above the threshold. Our forecasts will be updated annually to provide policy makers with advanced warning of the evolving probability and duration of future warming events.

Plain Language Summary The Paris Agreement calls for efforts to limit human-induced global warming to less than 1.5 °C above preindustrial levels. Observations of global mean temperature contain both human-induced temperature change and superimposed natural variability. Natural variability may temporarily add to the underlying human-induced warming, leading to observed temperatures that are higher than 1.5 °C for short-term periods. This would not necessarily exceed the Paris agreement, which is usually interpreted to refer to long-term averages, but would give an important indication that the threshold is being approached. If exceedance occurs, policy makers will require guidance regarding how long temperatures will remain above the threshold. Here we develop a new capability to predict the likelihood that global temperature will exceed 1.5 °C above preindustrial levels in the coming 5 years. We use decadal climate predictions that are regularly produced by several international climate prediction centers. Importantly, these predictions take into account the observed present day conditions since this is essential to predict the evolution of natural variability. For the period 2017 to 2021 we predict a 38% and 10% chance, respectively, of monthly or yearly temperatures exceeding 1.5 °C, with virtually no chance of the 5-year mean being above the threshold. We will update our forecasts every year to provide policy makers with advanced warning of the evolving probability and duration of future warming events.

1. Introduction

The Paris Agreement (UNFCCC, 2015) aims to limit global surface temperature rise to "well below 2 °C above preindustrial levels and to pursue efforts to limit the temperature increase to 1.5 °C above preindustrial levels." Although the precise meaning of exceeding a given temperature threshold was not defined, Rogelj

et al. (2017) interpret it as human-induced changes in climatological means, averaged over periods of 30 years. However, this will be difficult to verify with observations until well after the event, as natural internal variability may add to human-induced warming to produce temporary threshold exceedances in the observations earlier than would be expected from human influences (Henley & King, 2017; Joshi et al., 2011).

The Paris Agreement (UNFCCC, 2015) also recognizes "the importance of averting, minimizing and addressing loss and damage associated with the adverse effects of climate change, including extreme weather events" There is clear evidence that greater global mean warming increases the risk of extreme weather events (Seneviratne et al., 2016), impacting food security (Betts et al., 2018), temperature extremes, and hence mortality rates, in many regions including Europe (Dosio & Fischer, 2018; King & Karoly, 2017), Australia (King et al., 2017), China, and East Asia (Li et al., 2018; Shi et al., 2018), European storms and precipitation extremes (Barcikowska et al., 2018), and coral bleaching events in the Great Barrier Reef (King et al., 2017). Hence, even temporary excursions of global mean temperature above 1.5 °C, with associated increases in the risks of extreme weather events, are relevant for policy makers.

Recent record breaking temperatures (Hu & Fedorov, 2017; Su et al., 2017) highlight the proximity of the 1.5 °C target, and the possibility that it could be exceeded at least temporarily in the near future. If this occurs, policy makers will require guidance regarding the amount of human-induced warming (Haustein et al., 2017) and how long temperatures will remain above the target. Here we develop predictions of the likelihood that global mean temperature will exceed 1.5 °C for different length periods in the coming 5 years. These provide advanced warning of impending risks as well as guidance on the likely duration of exceedance events.

In order to capture natural internal variability, climate predictions must be initialized with the observed state of the climate system. Decadal climate prediction systems have recently been developed for this purpose and have been shown to improve predictions of global mean temperature relative to uninitialized projections (Boer et al., 2013; Doblas-Reyes et al., 2013; Fyfe et al., 2011; Smith et al., 2007). Furthermore, many international forecast centers are now routinely producing and exchanging initialized decadal predictions in near real-time (Smith, Scaife, et al., 2013). We therefore use the multimodel ensemble of initialized real-time decadal climate predictions provided by this international activity to develop and evaluate predictions of the probability of temporarily exceeding 1.5 °C above preindustrial levels. The intention is that these forecast probabilities will be updated annually as new forecasts become available, and made available to policy makers as part of the World Climate Research Program (WCRP) Grand Challenge on Near Term Climate Prediction [https://www.wcrp-climate.org/grand-challenges/gc-near-term-climate-prediction].

2. Data and Methods

2.1. Observations

Observations of global mean near surface temperature are taken as the average of HadCRUT4 (Morice et al., 2012), NASA-GISS (Hansen et al., 2010), and NCDC (Karl et al., 2015). The Paris Agreement (UNFCCC, 2015) did not define the period to be used as the preindustrial reference, and different choices will affect whether the 1.5 °C threshold has been exceeded or not (Schurer et al., 2017; Visser et al., 2018). In common with other studies (Collins et al., 2013; Henley & King, 2017) we take a pragmatic approach (Hawkins et al., 2017) and use the period 1850 to 1900 to represent preindustrial conditions since earlier periods suffer from a lack of direct observations. Estimates of warming for the period 1850–1900 relative to earlier periods range from 0.0 to 0.2 °C (Hawkins et al., 2017; Schurer et al., 2017), which would increase our predicted chance of a temporary excursion over 1.5 °C.

2.2. Models

We analyze global mean surface temperatures from 9 state-of-the-art decadal climate prediction systems (Table 1), all of which have contributed to the international activity to exchange decadal predictions in near real-time that has been ongoing since 2010 (Smith, Scaife, et al., 2013). Decadal climate predictions include the same external forcings (greenhouse gases, anthropogenic and volcanic aerosols, changes in solar irradiance) as centennial timescale projections of climate change (Collins et al., 2013) but are additionally initialized with observations of the state of the climate system. In this way, they are able to predict



Table 1

Summary of Initialized Decadal Forecasts

Name	Institute	Hindcast and forecast start dates	Model and resolution ^a atmosphere (A) ocean(O)	Ensemble size	References
BSC	Barcelona Supercomputing Center, Spain	1st November: 1960, 1961, 1962,, 2016	EC-Earth A: T255 L 62 O: nominal 1 L 46	5 (10 for the 2017 forecast)	Ménégoz et al. (2018), Doblas-Reyes et al. (2018)
CCCma	Canadian Centre for Climate Modelling and Analysis, Environment and Climate Change Canada	1st January: 1961, 1962, 1963,, 2018	CanCM4 A: T63 L35 O: 1.4 × 0.9 L40	10	Kharin et al. (2012)
GFDL	Geophysical Fluid Dynamics Laboratory, USA	1st January: 1961, 1962, 1963,, 2017	GFDL-CM2.1 A: $2.5 \times 2.0 L24$ O: nominal 1 L50	10	Yang et al. (2013)
MIROC5	University of Tokyo, National Institute for Environmental Studies, and Japan Agency for Marine-Earth Science and Technology, Japan	1st January: 1961, 1962, 1963,, 2017	MIROC5 A:T85 L40 O:1.4 × 1.4 L50	9	Chikamoto et al. (2012), Mochizuki et al. (2012)
МОНС	Met Office Hadley Centre, UK	1st November: 1960, 1962, 1965, 1968,, 2008, 2009, 2010,, 2016	HadGEM3-GC2 A: 0.8 × 0.6 L85 O: 0.25 × 0.25 L75	10 (20 for 2016 forecast)	Dunstone et al. (2016), Sheen et al. (2017)
MPI	Max Planck Institute for Meteorology, Germany	1st January: 1961, 1962, 1963,, 2017	MPI-ESM A: T63 L47 O: 1.5 × 1.5 L40	10	Pohlmann et al. (2013)
MRI	Meteorological Research Institute, Japan	1st January: 1961, 1966, 1971,, 2011, 2012, 2013, , 2017	MRI-CGCM3 A:1.125 × 1.125 L48 O: 1.0 × 0.46 L51	9	
NCAR	National Center for Atmospheric Research, USA	1st November: 1960, 1961, 1962,, 2015	CESM1.1 A: 1.25 × 0.9 L30 O: nominal 1 L60	10	Yeager et al. (2018)
SMHI	Swedish Meteorological and Hydrological Institute, Sweden	1st January: 1992, 1993, 1994,, 2009, 2016	EC-Earth A: T255 L 91 O: nominal 1 L 46	9	

^aDegrees longitude \times latitude, or spectral (T159 \approx 1.125°), L vertical levels.

some aspects of unforced internal variability, and may also improve the simulated response to external forcings.

We assess the likely forecast skill by evaluating hindcasts (retrospective forecasts made using data that would have been available at the time) covering the historical period since 1960. The hindcast start dates for the different systems are given in Table 1, and all available data are used to assess skill. Because decadal forecasts and hindcasts are expensive to produce, especially with high resolution climate models, some systems only extend to 5 years ahead. We therefore restrict our analysis to the first 5 years of each hindcast and forecast for all systems. We also compare with uninitialized climate model simulations from the 5th Coupled Model Intercomparison Project (K. E. Taylor et al., 2012) to assess the impact of initialization. For this we obtained monthly mean global mean surface temperature from 38 models (documented in Appendix A of Hawkins & Sutton, 2016) driven by historical and RCP4.5 radiative forcings.

Because models are imperfect, initialized forecasts drift away from the observations toward the biased model state (Boer et al., 2016; Gangstø et al., 2013; Hermanson et al., 2017; Smith, Eade, et al., 2013). For each individual decadal prediction system, a lead-time dependent drift has therefore been diagnosed from the hind-casts and removed from each month of the initialized model data to produce anomalies relative to the period 1971 to 2000. Each uninitialized simulation is similarly converted into anomalies relative to its average of the period 1971 to 2000 for each month. For both initialized and uninitialized models, values relative to preindustrial conditions are then obtained by adding the observed difference of 0.41 °C between the periods 1971 to 2000 and 1850 to 1900.

We compute the probability of a month (or year) temporarily exceeding 1.5 °C above preindustrial conditions as the fraction of ensemble members that have at least one monthly (or yearly) mean during the 5-year fore-cast period that exceeds this threshold.



Figure 1. Evaluation of prediction skill and reliability. (a) RMSE as a function of forecast lead time for predictions of annual mean global average temperature. Initialized forecasts (red) are compared with uninitialized simulations (blue, with bars indicating the 5–95% confidence interval where differences could have occurred by chance). (b) Reliability of probabilistic forecasts for exceeding annual mean thresholds defined by increases ranging from 0.25 (yellow) to 0.45 °C (dark red) relative to the annual mean temperature preceding the forecast start date. Exceeding 1.5 °C requires an increase of 0.36 °C relative to the annual mean temperature in 2016. Results are from the initialized hindcasts (Table 1). Bars show the 5% to 95% range of uncertainties in the observed frequencies. RMSE = root-mean-squared error.

2.3. Forecast Quality Measures

We assess the deterministic and probabilistic skill of the forecasts using root-mean-square error (RMSE) and reliability (e.g., Jolliffe & Stephenson, 2003). To assess the impact of initialization we generate 5,000 random samples (with replacement) of the differences in RMSE between initialized and uninitialized hindcasts and compute their 5% to 95% range. Samples are taken in blocks of 3 years to allow for serial autocorrelation.

We cannot directly assess the reliability of forecasts of the probability of exceeding 1.5 °C because this event has not yet occurred in the observations. Instead, we assess the reliability of probabilistic forecasts for exceeding a range of thresholds defined by increases relative to the annual mean temperature preceding the forecast start date. We evaluate increases ranging from 0.25 to 0.45 °C, noting that exceeding 1.5 °C requires an increase of 0.36 °C relative to the annual mean temperature in 2016. Uncertainties in the observed frequencies of occurrence are estimated by computing the 5% to 95% range from 5,000 random samples (with replacement) of the observations. We plot the average of the forecast probabilities within each bin rather than the central bin value to reduce biases in the reliability diagram (Bröcker & Smith, 2007).

3. Results

A positive impact of initialization on the skill of predicting global mean surface temperature has been demonstrated in previous studies (Boer et al., 2013; Doblas-Reyes et al., 2013; Fyfe et al., 2011; Smith et al., 2007) and is reproduced in our results which show lower RMSE in initialized compared to uninitialized forecasts for all of the 5 years of the forecasts, with improvements significant at the 95% level in each of the first 4 years (Figure 1a). In addition, in the initialized ensemble, the forecast probability for exceeding a range of thresholds is generally consistent with the observed frequency of occurrence (Figure 1b). There is a suggestion that forecasts of medium probabilities (e.g., around 30% to 40%) may be slightly overconfident (i.e., forecast probabilities higher than observed frequencies). However, the uncertainties are large and intersect the diagonal (representing perfect reliability) for at least one of the thresholds for all of the probability bins. Overall, the hindcasts show that initialized forecasts are more accurate than uninitialized projections for the coming 5 years and provide reasonably reliable probabilities of exceeding warming thresholds.

Time series of rolling 5-year average global mean surface temperature for the observations along with the initialized and uninitialized hindcasts and forecasts are shown in Figure 2. Consistent with the lower RMSEs (Figure 1a), the initialized hindcasts are generally closer to the observations than are the uninitialized simulations. This is especially clear during the 1998 to 2012 slowdown in the rate of global surface temperature warming (e.g., Fyfe et al., 2016). However, although the initialized hindcasts are cooler than the uninitialized simulations during this period (also noted by Doblas-Reyes et al., 2013, and Guemas et al., 2013), their mean is warmer than the observations, although the observations lie well within the spread of the predictions. The differences between models and observations are caused by a combination of unpredicted internal



Figure 2. Predicted chance of temporarily exceeding 1.5 °C warming above preindustrial conditions. Time series of rolling 5-year average global mean surface temperature relative to preindustrial (left axis) in observations (black) compared to initialized forecasts (red, solid line showing the ensemble mean and shading the 5–95% ensemble range) and uninitialized simulations (blue, solid line showing the ensemble mean and shading the 5–95% ensemble range). Green shading shows the probability (right axis) from the initialized forecasts of temporarily exceeding 1.5 °C for at least 1 month (light green) and at least 1 year (dark green) during the 5-year forecast period. The date represents the start of each 5-year forecast period.

variability, forcings that are not included in the model simulations (e.g., the early 2000s volcanic eruptions, Santer et al., 2014), and imperfect simulations of external forcings (e.g., Fyfe et al., 2016; Smith et al., 2016). However, recent record breaking global temperatures suggest that the surface warming slowdown is now over (Hu & Fedorov, 2017; Su et al., 2017).

The hindcast and forecast probabilities of temporarily exceeding 1.5 °C are also shown in Figure 2 (green shading). Before 1998 there was virtually no chance that a month or year would exceed 1.5 °C, but the chance has been increasing over the last two decades. In the latest forecast the chances of a month, year, or 5 years exceeding 1.5 °C during the 5-year period 2017 to 2021 are 38%, 10%, and 0%, respectively. For comparison, the uninitialized projections yield probabilities of 89%, 50%, and 23%, respectively, for a month, year, and 5 years (not shown), but these are warmer than the initialized predictions and the observations in recent periods (Figure 2). The forecast probabilities are expected to increase in future: based on the current model trend of 0.2 °C per decade, it is more likely than not that a month will exceed 1.5 °C in the period 2020 to 2024.

It is well recognized that there are different pathways for global warming to reach the thresholds stated in the Paris Agreement (e.g., Goodwin et al., 2018). To investigate the different possible mechanisms through which the climate might temporarily exceed 1.5 °C, we show in Figure 3 the monthly patterns of near surface temperature at the time of maximum global temperature for every ensemble member that exceeds 1.5 °C in the latest forecast from the Met Office decadal prediction system (Met Office Hadley Centre, Table 1). It is well established that global mean temperature is increased by heat released from the tropical Pacific during the positive phase of the El Niño Southern Oscillation (ENSO; e.g., Lean & Rind, 2008; Trenberth et al., 2002). Hence, many of the months show a positive ENSO, characterized by an anomalously warm (cool) eastern (western) tropical Pacific. High global temperatures are also associated with a weakened Southern Annular Mode characterized by high mean sea level pressure (not shown) and high temperatures over Antarctica (Figure 3I), but this link may not be causal because a weaker Southern Annular Mode may itself be driven by ENSO (Fogt & Bromwich, 2006; L'Heureux & Thompson, 2006; Wang & Cai, 2013). Increased global temperatures are also associated with a warm Eurasia (Figure 3I) driven by a positive Arctic Oscillation (AO, not shown), consistent with similar links seen on decadal timescales (Delworth et al., 2016; Iles & Hegerl, 2017). This link is likely to be independent of ENSO which tends to promote a negative AO (e.g., Brönnimann et al., 2007; Ineson & Scaife, 2009). Analysis of CanCM4 provides similar results (not shown) although the link with a positive AO is less clear in this model. Overall, our results suggest that temporary excursions above 1.5 °C are associated with positive phases of ENSO and/or the AO. The first exceedances of 1.5 °C are therefore most likely to occur during the boreal winter and spring since both ENSO and the AO have the largest variance and impact in these seasons.



Figure 3. Patterns associated with temporarily exceeding 1.5 °C. (a)–(k) Monthly mean near surface temperature (plotted as standard deviations of the ensemble spread relative to the ensemble mean at each grid point) at the time of maximum global mean temperature for each Met Office Hadley Centre ensemble member that exceeds 1.5 °C in the period 2017 to 2021. The dates are indicated for each month, and their average is shown in the bottom right panel (I). Note that the same date (e.g. April 2021) can occur by chance in different ensemble members.

4. Conclusions

We have developed a new capability to predict the likelihood that global mean temperature will temporarily exceed 1.5 °C above preindustrial levels within the coming 5 years. The forecast utilizes an ongoing international activity to produce a multimodel ensemble of near real-time decadal climate predictions (Smith, Scaife, et al., 2013). In agreement with previous studies we found that these initialized decadal predictions provide more accurate forecasts of global mean temperature than uninitialized climate model projections. We also found that forecast probabilities of exceeding a range of thresholds are reasonably reliable.

For the 5-year period 2017 to 2021 our forecast indicates a 38% and 10% chance, respectively, that an individual month or year will exceed 1.5 °C, with virtually no chance of the 5-year mean being above the threshold. In contrast, the uninitialized model projections show much higher probabilities of 89%, 50%, and 23%, respectively, for a month, year, or the 5-year period. The lower probabilities from the initialized predictions are consistent with improved predictions during the slowdown in surface temperature warming in the early-2000s. However, we note that the initialized predictions were still warmer than observations during

10C

the slowdown period and may have also overestimated the probability of exceeding 1.5 °C. Whether this is predominantly caused simply by internal variability, or whether there are errors in forcings or simulated internal variability (e.g., Fyfe et al., 2016; Medhaug et al., 2017), or incorrect model responses to aerosols (Smith et al., 2016), is not fully resolved. Indeed, while decadal climate prediction systems improve near-term predictions compared to uninitialized projections, they do not avoid the need for a fuller understanding of short-term variations in forcings and responses. In particular, modeled circulation responses to several factors including volcanoes, ozone, and solar variability, may be too weak (Scaife & Smith, 2018), and the impacts of stratospheric water vapor (Solomon et al., 2010) on decadal climate are unresolved. The slowdown now appears to be over (Hu & Fedorov, 2017; Su et al., 2017) and our forecast probabilities could be too low in the event of accelerated warming caused either by an unpredicted reversal of internal variability or incorrectly simulated aerosol-induced changes. Conversely, a major volcanic eruption (which is unpredictable) would cool global temperature for several years and render our probabilities too high (Illing et al., 2018; Timmreck et al., 2016).

Our results suggest that temporary excursions above 1.5 $^{\circ}$ C are most likely caused by positive ENSO and/or AO events. Regional patterns of warming, and hence associated climate impacts, are therefore likely to be very different for temporary excursions than they would be if long-term global temperatures exceed 1.5 $^{\circ}$ C. In our predictions, short-term warming is also most likely to occur during the boreal winter and spring when ENSO and the AO have the largest variability.

We stress that a temporary excursions over 1.5 °C above preindustrial conditions would not necessarily constitute an exceedance of the Paris Agreement, which has been interpreted to refer to human-induced climate change over periods of at 30 years (Rogelj et al., 2017). Indeed, complying with the Paris Agreement is still achievable (Rogelj et al., 2018) and important, given the impact of further warming on Arctic Sea Ice (Jahn, 2018; Niederdrenk & Notz, 2018; Sigmond et al., 2018), Greenland, and West Antarctic ice sheets and sea level rise (Rasmussen et al., 2018), regional climate (Jacob et al., 2018; M. A. Taylor et al., 2018), and extremes (Dosio & Fischer, 2018; Li et al., 2018). The first temporary excursions above 1.5 °C, if they occur, would not necessarily lead to many of these effects but they would be an important indication of the proximity to the Paris threshold and could generate interest from the media and general public. Our forecasts will therefore be updated annually to provide advanced warning of the likelihood and duration of such events.

References

- Barcikowska, M. J., Weaver, S. J., Feser, F., Russo, S., Schenk, F., Stone, D. A., et al. (2018). Euro-Atlantic winter storminess and precipitation extremes under 1.5 °C vs. 2 °C warming scenarios. *Earth System Dynamics*, *9*, 679–699. https://doi.org/10.5194/esd-9-679-2018
- Betts, R. A., Alfieri, L., Bradshaw, C., Caesar, J., Feyen, L., Friedlingstein, P., et al. (2018). Changes in climate extremes, fresh water availability and vulnerability to food insecurity projected at 1.5 °C and 2 °C global warming with a higher-resolution global climate model. *Philosophical Transactions of the Royal Society A*, 376. https://doi.org/10.1098/rsta.2016.0452
- Boer, G. J., Kharin, V. V., & Merryfield, W. J. (2013). Decadal predictability and forecast skill. Climate Dynamics, 41, 1817–1833. https://doi.org/ 10.1007/s00382-013-1705-0
- Boer, G. J., Smith, D. M., Cassou, C., Doblas-Reyes, F., Danabasoglu, G., Kirtman, B., et al. (2016). The decadal climate prediction project (DCPP) contribution to CMIP6. *Geoscientific Model Development Discussion*, 1–32. https://doi.org/10.5194/gmd-2016-78
- Bröcker, J., & Smith, L. A. (2007). Increasing the reliability of reliability diagrams. Weather and Forecasting, 22, 651–661. https://doi.org/ 10.1175/WAF993.1
- Brönnimann, S., Xoplaki, E., Casty, C., Pauling, A., & Luterbacher, J. (2007). ENSO influence on Europe during the last centuries. *Climate Dynamics*, 28, 181–197.
- Chikamoto, Y., Kimoto, M., Ishii, M., Mochizuki, T., Sakamoto, T. T., Tatebe, H., et al. (2012). An overview of decadal climate predictability in a multi-model ensemble by climate model MIROC. *Climate Dynamics*, 40, 1201–1222.
- Collins, M., et al. (2013). Long-term climate change: Projections, commitments and irreversibility. In T. F. Stocker, et al. (Eds.), *Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change* (pp. 1029–1136). Cambridge, UK, and New York: Cambridge University Press. https://doi.org/10.1017/CBO9781107415324.024
- Delworth, T. L., Zeng, F., Vecchi, G. A., Yang, X., Zhang, L., & Zhang, R. (2016). The North Atlantic Oscillation as a driver of rapid climate change in the Northern Hemisphere. *Nature Geoscience*, 9, 509–512. https://doi.org/10.1038/NGE02738
- Doblas-Reyes, F. J., Acosta Navarro, J. C., Acosta, M., Batté, L., Bellprat, O., Bilbao, R., et al. (2018). Using EC-Earth for climate prediction research, ECMWF Newsletter.
- Doblas-Reyes, F. J., Andreu-Burillo, I., Chikamoto, Y., García-Serrano, J., Guemas, V., Kimoto, M., et al. (2013). Initialized near-term regional climate change prediction. *Nature Communications*, *4*, 1715. https://doi.org/10.1038/ncomms2704
- Dosio, A., & Fischer, E. M. (2018). Will half a degree make a difference? Robust projections of indices of mean and extreme climate in Europe under 1.5°C, 2°C, and 3°C global warming. *Geophysical Research Letters*, *45*, 935–944. https://doi.org/10.1002/2017GL076222
- Dunstone, N. J., Smith, D. M., Scaife, A. A., Hermanson, L., Eade, R., Robinson, N., et al. (2016). Skilful predictions of the winter North Atlantic Oscillation one year ahead. *Nature Geoscience*, *9*, 809–814. https://doi.org/10.1038/NGEO2824
- Fogt, R. L., & Bromwich, D. H. (2006). Decadal variability of the ENSO teleconnection to the high-latitude South Pacific governed by coupling with the southern annular mode. *Journal of Climate*, 19, 979–997. https://doi.org/10.1175/JCLI3671.1

Acknowledgments

D.M.S., A.A.S., N.J.D., L.H., and R.E. were supported by the Met Office Hadley Centre Climate Programme funded by BEIS and Defra and by the European Commission Horizon 2020 EUCP project (GA 776613). R.B., L.P.C., F.J.D.R., and M. M. were supported by the H2020 EUCP (GA 776613) and the Spanish MINECO CLINSA (CGL2017-85791-R) and HIATUS (CGL2015-70353-R) projects. L.P.C.'s contract is cofinanced by the MINECO under Juan de la Cierva Incorporacin postdoctoral fellowship number IJCI-2015-23367. W.A.M. and H.P. acknowledge funding from the German Federal Ministry for Education and Research (BMBF) project MiKlip (FKZ 01LP1519A). The NCAR contribution was supported by the US National Oceanic and Atmospheric Administration (NOAA) Climate Program Office under Climate Variability and Predictability Program grant NA13OAR4310138, by the US National Science Foundation (NSF) Collaborative Research EaSM2 grant OCE-1243015, by the Regional and Global Climate Modeling Program (RGCM) of the US Department of Energy's, Office of Science (BER), Cooperative Agreement DE-FC02 97ER62402, and by the NSF through its sponsorship of NCAR. The NCAR simulations were generated using computational resources provided by the National Energy Research Scientific Computing Center, which is supported by the Office of Science of the U.S. Department of Energy under Contract DE-AC02-05CH11231, as well as by an Accelerated Scientific Discovery grant for Cheyenne that was awarded by NCAR's Computational and Information Systems Laboratory. The EC-EARTH simulations by SMHI were performed on resources provided by the Swedish National Infrastructure for Computing (SNIC) at NSC. Data used to create the figures are available at 10.5281/ zenodo.1434700.

Fyfe, J. C., Meehl, G. A., England, M. H., Mann, M. E., Santer, B. D., Flato, G. M., et al. (2016). Making sense of the early-2000s warming slowdown. Nature Climate Change, 6, 224–228. https://doi.org/10.1038/nclimate2938

Fyfe, J. C., Merryfield, W. J., Kharin, V., Boer, G. J., Lee, W.-S., & von Salzen, K. (2011). Skillful predictions of decadal trends in global mean surface temperature. *Geophysical Research Letters*, 38, L22801. https://doi.org/10.1029/2011GL049508

Gangstø, R., Weigel, A. P., Liniger, M. A., & Appenzeller, C. (2013). Methodological aspects of the validation of decadal predictions. *Climate Research*, *55*, 181–200. https://doi.org/10.3354/cr01135

Goodwin, P., Katavouta, A., Roussenov, V. M., Foster, G. L., Rohling, E. J., & Williams, R. G. (2018). Pathways to 1.5 °C and 2 °C warming based on observational and geological constraints. *Nature Geoscience*, *11*(2), 102–107. https://doi.org/10.1038/s41561-017-0054-8

Guemas, V., Doblas-Reyes, F. J., Andreu-Burillo, I., & Asif, M. (2013). Retrospective prediction of the global warming slowdown in the past decade. *Nature Climate Change*, 3, 649–653. https://doi.org/10.1038/nclimate1863

Haustein, K., Allen, M. R., Forster, P. M., Otto, F. E. L., Mitchell, D. M., Matthews, H. D., & et al. (2017). A real-time global warming index. Scientific Reports, 7, 15417. https://doi.org/10.1038/s41598-017-14828-5

Hawkins, E., et al. (2017), Estimating change in global temperature since the pre-industrial period, Bulletin of the American Meteorological Society, 98(9), 1841–1856, doi:https://doi.org/10.1175/BAMS-D-16-0007.1.

Hawkins, E., & Sutton, R. (2016). Connecting climate model projections of global temperature change with the real world. Bulletin of the American Meteorological Society, 97, 963–980. https://doi.org/10.1175/BAMS-D-14-00154.1

Henley, B. J., & King, A. D. (2017). Trajectories toward the 1.5 °C Paris target: Modulation by the Interdecadal Pacific Oscillation. *Geophysical Research Letters*, 44, 4256–4262. https://doi.org/10.1002/2017GL073480

Hermanson, L., Ren, H. L., Vellinga, M., Dunstone, N. D., Hyder, P., Ineson, S., et al. (2017). Different types of drifts in two seasonal forecast systems and their dependence on ENSO. *Climate Dynamics*, *51*, 1411–1426. https://doi.org/10.1007/s00382-017-3962-9

Hu, S., & Fedorov, A. V. (2017). The extreme El Niño of 2015-2016 and the end of global warming hiatus. *Geophysical Research Letters*, 44, 3816–3824. https://doi.org/10.1002/2017GL072908

Illes, C., & Hegerl, G. (2017). Role of the North Atlantic Oscillation in decadal temperature trends. Environmental Research Letters, 12, 114010. https://doi.org/10.1088/1748-9326/aa9152

Illing, S., Kadow, C., Pohlmann, H., & Timmreck, C. (2018). Assessing the impact of a future volcanic eruption on decadal predictions. Earth System Dynamics Discussions, 1–26. https://doi.org/10.5194/esd-2018-5

Ineson, S., & Scaife, A. A. (2009). The role of the stratosphere in the European climate response to El Niño. Nature Geoscience, 2, 32–36. https:// doi.org/10.1038/ngeo381

Jacob, D., Kotova, L., Teichmann, C., Sobolowski, S. P., Vautard, R., Donnelly, C., et al. (2018). Climate impacts in Europe under +1.5 °C global warming. *Earth's Future, 6*, 264–285. https://doi.org/10.1002/2017EF000710

Jahn, A. (2018), Reduced probability of ice-free summers for 1.5C compared to 2C warming, nature climate change, doi:https://doi.org/ 10.1038/s41558-018-0127-8

Jolliffe, I. T., & Stephenson, D. B. (Eds) (2003). Forecast verification: A practitioner's guide in atmospheric science. London: Wiley and Sons.

Joshi, M., Hawkins, E., Sutton, R., Lowe, J., & Frame, D. (2011). Projections of when temperature change will exceed 2 °C above pre-industrial levels. *Nature Climate Change*, 1, 407–412. https://doi.org/10.1038/nclimate1261

Karl, T. R., Arguez, A., Huang, B., Lawrimore, J. H., McMahon, J. R., Menne, M. J., et al. (2015). Possible artifacts of data biases in the recent global surface warming hiatus. *Science*, 348, 1469–1472.

Kharin, V. V., Boer, G. J., Merryfield, W. J., Scinocca, J. F., & Lee, W.-S. (2012). Statistical adjustment of decadal predictions in a changing climate. Geophysical Research Letters, 39, L19705. https://doi.org/10.1029/2012GL052647

King, A. D., & Karoly, D. (2017). Climate extremes in Europe at 1.5 and 2 degrees of global warming. *Environmental Research Letters*, 12, 114,031. https://doi.org/10.1088/1748-9326/aa8e2c

King, A. D., Karoly, D. J., & Henley, B. J. (2017). Australian climate extremes at 1.5 °C and 2 °C of global warming. *Nature Climate Change*, 7, 412–416. https://doi.org/10.1038/nclimate3296

Lean, J. L., & Rind, D. H. (2008). How natural and anthropogenic influences alter global and regional surface temperatures: 1889 to 2006. Geophysical Research Letters, 35, L18701. https://doi.org/10.1029/2008GL034864

L'Heureux, M. L., & Thompson, W. J. (2006). Observed relationships between the El Niño-Southern Oscillation and the extratropical zonalmean circulation. *Journal of Climate*, 19, 276–287. https://doi.org/10.1175/JCLI3617.1

Li, D., Zhou, T., Zou, L., Zhang, W., & Zhang, L. (2018). Extreme high-temperature events over East Asia in 1.5 °C and 2 °C warmer futures:

Analysis of NCAR CESM low-warming experiments. *Geophysical Research Letters*, 45, 1541–1550. https://doi.org/10.1002/2017GL076753 Medhaug, I., Stolpe, M. B., Fischer, E. M., & Knutti, R. (2017). Reconciling controversies about the 'global warming hiatus'. *Nature*, 545, 41–47. https://doi.org/10.1038/nature22315

Ménégoz, M., Bilbao, R., Bellprat, O., Guemas, V., & Doblas-Reyes, F. J. (2018). Forecasting the climate response to volcanic eruptions: Prediction skill related to stratospheric aerosol forcing. *Environmental Research Letters*, 13, 064022. https://doi.org/10.1088/1748-9326/ aac4db

Mochizuki, T., Kimoto, M., Ishii, M., Chikamoto, Y., Tatebe, H., Komuro, Y., et al. (2012). Decadal prediction using a recent series of MIROC global climate models. *Journal of the Meteorological Society of Japan*, 90A, 373–383. https://doi.org/10.2151/jmsj.2012-A22

Morice, C. P., Kennedy, J. J., Rayner, N. A., & Jones, P. D. (2012). Quantifying uncertainties in global and regional temperature change using an ensemble of observational estimates: The HadCRUT4 data set. *Journal of Geophysical Research*, 117, D08101. https://doi.org/10.1029/ 2011JD017187

Niederdrenk, A. L., & Notz, D. (2018). Arctic sea ice in a 1.5 °C warmer world. Geophysical Research Letters, 45, 1963–1971. https://doi.org/ 10.1002/2017GL076159

Pohlmann, H., Müller, W. A., Kulkarni, K., Kameswarrao, M., Matei, D., Vamborg, F. S. E., et al. (2013). Improved forecast skill in the tropics in the new MiKlip decadal climate predictions. *Geophysical Research Letters*, 40, 5798–5802. https://doi.org/10.1002/2013GL058051

Rasmussen, D. J., Bittermann, K., Buchanan, M. K., Kulp, S., Strauss, B. H., Kopp, R. E., & et al. (2018). Extreme sea level implications of 1.5 °C, 2.0 °C, and 2.5 °C temperature stabilization targets in the 21st and 22nd centuries. *Environmental Research Letters*, *13*, 034040. https://doi. org/10.1088/1748-9326/aaac87

Rogelj, J., Popp, A., Calvin, K. V., Luderer, G., Emmerling, J., Gernaat, D., et al. (2018). Scenarios towards limiting global mean temperature increase below 1.5 °C. *Nature Climate Change*, 8, 325–332. https://doi.org/10.1038/s41558-018-0091-3

Rogelj, J., Schleussner, C.-F., & Hare, W. (2017). Getting it right matters: Temperature goal interpretations in geoscience research. Geophysical Research Letters, 44, 10662–10665. https://doi.org/10.1002/2017GL075612

Hansen, J., Ruedy, R., Sato, M., & Lo, K. (2010). Global surface temperature change. *Reviews of Geophysics*, 104, RG4004. https://doi.org/ 10.1029/1999JD900835



Santer, B. D., Bonfils, C., Painter, J. F., Zelinka, M. D., Mears, C., Solomon, S., et al. (2014). Volcanic contributions to decadal changes in tropospheric temperature. *Nature Geoscience*, 7, 185–189. https://doi.org/10.1038/ngeo2098

Scaife, A. A., & Smith, D. M. (2018). A signal-to-noise paradox in climate science. Nature Climate and Atmospheric Science, 1. https://doi.org/ 10.1038/s41612-018-0038-4

Schurer, A. P., Mann, M. E., Hawkins, E., Tett, S. F. B., & Hegerl, G. C. (2017). Importance of the pre-industrial baseline for likelihood of exceeding Paris goals. *Nature Climate Change*, 7, 563–567. https://doi.org/10.1038/NCLIMATE3345

Seneviratne, S. I., M. G. Donat, A, J. Pitman, R. Knutti and R. L. Wilby (2016), Allowable CO emissions based on regional and impact-related climate targets, *Nature*, 529, 477–483, doi: https://doi.org/10.1038/nature16542

Sheen, K. L., Smith, D. M., Dunstone, N. J., Eade, R., Rowell, D. P., & Vellinga, M. (2017). Skilful prediction of Sahel summer rainfall on interannual and multi-year timescales. *Nature Comms*, 8, 14,966. https://doi.org/10.1038/ncomms14966

Shi, C., Jiang, Z.-H., Chen, W.-L., & Li, L. (2018). Changes in temperature extremes over China under 1.5 °C and 2 °C global warming targets. *Advances in Climate Change Research*, *9*, 120–129. https://doi.org/10.1016/j.accre.2017.11.003

Sigmond, M., Fyfe, J. C., & Swart, N. C. (2018). Ice-free Arctic projections under the Paris Agreement. *Nature Climate Change*, 8, 404–408. https://doi.org/10.1038/s41558-018-0124-y

Smith, D. M., Booth, B. B. B., Dunstone, N. J., Eade, R., Hermanson, L., Jones, G. S., et al. (2016). Role of volcanic and anthropogenic aerosols in recent slowdown in global surface warming. *Nature Climate Change*, *6*, 936–940. https://doi.org/10.1038/NCLIMATE3058

Smith, D. M., Cusack, S., Colman, A. W., Folland, C. K., Harris, G. R., & Murphy, J. M. (2007). Improved surface temperature prediction for the coming decade from a global climate model. *Science*, 317, 796–799. https://doi.org/10.1126/science.1139540

Smith, D. M., Eade, R., & Pohlmann, H. (2013). A comparison of full-field and anomaly initialization for seasonal to decadal climate prediction. *Climate Dynamics*, 41, 3325–3338. https://doi.org/10.1007/s00382-013-1683-2

Smith, D. M., Scaife, A. A., Boer, G. J., Caian, M., Doblas-Reyes, F. J., Guemas, V., et al. (2013). Real-time multi-model decadal climate predictions. *Climate Dynamics*, 41, 2875–2888. https://doi.org/10.1007/s00382-012-1600-0

Solomon, S., Rosenlof, K. H., Portmann, R. W., Daniel, J. S., Davis, S. M., Sanford, T. J., & et al. (2010). Contributions of stratospheric water vapor to decadal changes in the rate of global warming. *Science*, 327, 1219–1223. https://doi.org/10.1126/science.1182488

Su, I., Zhang, R., & Wang, H. (2017). Consecutive record-breaking high temperatures marked the handover from hiatus to accelerated warming. *Scientific Reports*, 7. https://doi.org/10.1038/srep43735

Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93, 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1

Taylor, M. A., Clarke, L. A., Centella, A., Bezanilla, A., Stephenson, T. S., Jones, J. J., et al. (2018). Future Caribbean climates in a world of rising temperatures: The 1.5 vs 2.0 dilemma. Journal of Climate, 31, 2907–2926. https://doi.org/10.1175/JCLI-D-17-0074.1

Timmreck, C., Pohlmann, H., Illing, S., & Kadow, C. (2016). The impact of stratospheric volcanic aerosol on decadal-scale climate predictions. Geophysical Research Letters, 43, 834–842. https://doi.org/10.1002/2015GL067431

Trenberth, K. E., Caron, J. M., Stepaniak, D. P., & Worley, S. (2002). Evolution of El Niño-Southern Oscillation and global atmospheric surface temperatures. *Journal of Geophysical Research*, 107(D8), 4065. https://doi.org/10.1029/2000JD000298

UNFCCC (2015). Adoption of the Paris Agreement. FCCC/CP/2015/L.9/Rev.1, 32 pp. Retrieved from https://unfccc.int/files/meetings/paris_ nov_2015/application/pdf/paris_agreement_english_pdf

Visser, H., Dangendorf, S., van Vuuren, D. P., Bregman, B., & Petersen, A. C. (2018). Signal detection in global mean temperatures after "Paris": An uncertainty and sensitivity analysis. *Climate of the Past*, 14(2), 139–155. https://doi.org/10.5194/cp-14-139-2018

Wang, G., & Cai, W. (2013). Climate-change impact on the 20th-century relationship between the southern annular mode and global mean temperature. *Scientific Reports*, *3*, 2039. https://doi.org/10.1038/srep02039

Yang, X., Rosati, A., Zhang, S., Delworth, T. L., Gudgel, R. G., Zhang, R., et al. (2013). A predictable AMO-like pattern in GFDL's fully-coupled ensemble initialization and decadal forecasting system. *Journal of Climate*, *26*(2), 650–661. https://doi.org/10.1175/JCLI-D-12-00231.1

Yeager, S. G., Danabasoglu, G., Rosenbloom, N., Strand, W., Bates, S., Meehl, G., et al. (2018). Predicting near-term changes in the Earth System: A large ensemble of initialized decadal prediction simulations using the Community Earth System Model. *Bulletin of the American Meteorological Society*, 99, 1867–1886. https://doi.org/10.1175/BAMS-D-17-0098.1